Author's response to reviews

Title: Identification of gene-based biomarkers of alpha particle radiation exposure in human blood cells

Authors:

Vinita Chauhan (vinita.chauhan@hc-sc.gc.ca)
Matthew Howland (matthew.howland@hc-sc.gc.ca)
Ruth Wilkins (ruth.wilkins@hc-sc.gc.ca)

Version: 3
Date: 22 January 2014

Author's response to reviews: see over
January 10, 2014

Dr Vinita Chauhan
Health Canada
775 Brookfield Rd.
Ottawa, Ontario
K1A 0K9
Tel. (613) 941-8516
Fax (613) 941-1734

Dear Editors

Please find enclosed a copy of our scientific paper entitled “Identification of gene-based biomarkers of alpha particle radiation exposure in human blood cells”. We would like to submit this paper to BM C Genomics for publication as it pertains to genomics in relation to radiation biodosimetry and is original work in this field of research. Submission of this manuscript has been approved by all authors and this manuscript is not currently submitted elsewhere for review or publication.

Thank you for your consideration,

Sincerely,

Vinita Chauhan, PhD.
Research Scientist